SS 04 Basic Module (4 SWS) with Tutorial/Übungen (2 SWS)

Frank Richter:

Grammatikformalismen für die Computerlinguistik

Seminar: Monday 9ct–11 at the SfS, Seminarraum 1.13 and Wednesday 9ct–11 at the SfS, Seminarraum 1.13 Seminar starts: Monday, 26. April 2004 Tutorial: Wednesday 11ct–13, Monika Toth, Hörsaal 0.02 Tutorial starts: Wednesday, 28. April 2004 Office Hours: Monday 11.00 – 12.00 Office No. at SfS Phone # Email 2.17 29-78489 fr@sfs.uni-tuebingen.de

• Webpage:

http://www.sfs.uni-tuebingen.de/∼fr/

• 9 Credits for:

– Obligatory attendance of class meetings – Evaluation:

∗ 50%: exercise sheets ∗ 50%: final exam • Final exam: in the last week of classes, probably July 21st.